Author's response to reviews

Title: Is health literacy related to health behaviors and cell phone usage patterns among the text4baby target population? A sample survey in 2 WIC clinics

Authors:

  Elisabeth A Poorman (elisabeth.poorman@gmail.com)
  Julie A Gazmararian (jgazma@emory.edu)
  Lisa K Elon (jelon@emory.edu)
  Ruth M Parker (rpark01@emory.edu)

Version: 4 Date: 17 February 2014

Author's response to reviews: see over
We would like to thank you for these final formatting comments. The title has been changed to “Is health literacy related to health behaviors and cell phone usage patterns among the text4baby target population? A sample survey in 2 metro Atlanta Women, Infant and Children (WIC) clinics.” The sentence postpartum OR were currently pregnant” has been changed to “postpartum or were currently pregnant.” With respect to variable border choices, we offer two versions: one with all borders for consistency, and one with variable borders, which may be easier to read. We leave it up to your judgment.

We hope that our responses are acceptable. Please contact me if further clarification is needed. We look forward to your evaluation of this resubmission.

Elisabeth Poorman, MD/MPH